Author’s response to reviews

Title: Intensified household contact tracing, prevention and treatment support versus enhanced standard of care for contacts of tuberculosis cases in South Africa: Study protocol for a household cluster-randomised trial

Authors:

Peter MacPherson (peter.macpherson@lstmed.ac.uk)
Emily Webb (emily.webb@lshtm.ac.uk)
Ebrahim Variava (variava@wordlonline.co.za)
Sanjay Lala (sanjay.lala@wits.ac.za)
Minja Milovanovic (milovanovicm@phru.co.za)
Andrew Ratsela (Rampule.Ratsela@ul.ac.za)
Limakatso Lebina (lebinal@phru.co.za)
Anthony Kinghorn (kinghorna@phru.co.za)
Neil Martinson (martinson@phru.co.za)

Version: 1 Date: 19 Sep 2019

Author’s response to reviews:

Dear Oliver Cocks,

RE: INFD-D-19-00112R1

Many thanks for your email regarding this manuscript. We have made the changes requested, and uploaded a clean version here.

You asked whether the maps map depicted in figure 1 are our own, or taken from another source. We sourced maps from the Database of Global Administrative Areas (GADM - https://gadm.org). GADM state that are "The data are freely available for academic use and other non-commercial use."
Please let us know if you have any further questions.

Kind regards,

Peter MacPherson (on behalf of the authors)